Author's response to reviews

Title: Lessons and implications from a mass immunization campaign in squatter settlements of Karachi, Pakistan: an experience from a cluster-randomized double-blinded controlled trial [NCT00125047]

Authors:

Mohammad I Khan (mohammad.imran@aku.edu)
Rion L Ochiai (rlochiai@ivi.int)
Hasan B Hamza (hasan.hamza@aku.edu)
Shah M Sahito (dr_sm1@hotmail.com)
Muhammad A Habib (atif.habib@aku.edu)
Naveed S Bhutto (naveed.bhutto@aku.edu)
Shahid Rasool (shahid.rasool@aku.edu)
Mahesh K Puri (mkpuri@ivi.int)
Mohammad Ali (mali@ivi.int)
Shafi M Wasan (mik@akunet.org)
Mohammad J Khan (jawed.khan@aku.edu)
Sajid B Soofi (sajid.soofi@aku.edu)
Remon A Elyazeed (Remon.Abu-Elyazeed@gsk.com)
Bernard Ivanoff (bivan@bluewin.ch)
Shaikh Q Nizami (qamaruddin.nizami@aku.edu)
Claudia M Galindo (claudiagalindo2004@yahoo.com)
Lorenz VON Seidlein (lseidlein@ivi.int)
Tikki Pang (pangt@who.ch)
Allan Donner (allan.donner@fmd.uwo.ca)
John D Clemens (jclemens@ivi.int)
Zulfiqar A Bhutta (zulfiqar.bhutta@aku.edu)

Version: 5 Date: 17 May 2006

Author's response to reviews: see over
Dear Editors,

On behalf of the authors I feel pleasure in thanking the reviewers and yourself for providing this opportunity to publish our work in *Trials*. We appreciate the efforts that were put in during the review process of our article. We believe the quality of the manuscript has improved a lot and would be more beneficial for the readers and those involved in future field trials in developing countries.

We have further incorporated your editorial suggestion in the manuscript and hereby submit as the final version of the paper and understand that no changes will be made afterwards.

Please feel free to contact if you have any questions.

Sincerely,

Zulfiqar A Bhutta
Department of Pediatrics, Aga Khan University & Hospital
P.O. Box No.3500
Karachi, Pakistan
Telephone: (92)-(21- 4930051) Ext: 4721
Telefax: (92)-(21- 4934294)
e-mail: zulfiqar.bhutta@aku.edu